Ulm, Germany, may 11nd, 2005

Dear BMC Editorial Team,

enclosed please find the third revision II of our manuscript entitled: "Effects of Mycophenolate Mofetil on Key Pattern of Coronary Restenosis: a Cascade of in vitro and ex vivo Models".

Please find here the point by point descriptions of the changes made:

1. Sentence case was used in the title.
2. Conclusions was added instead of Discussion in the abstract.

We hope that the manuscript can now be accepted for publication in BMC.

With best wishes

Rainer Voisard, MD